Author’s response to reviews

Title: Bradycardia and syncope as sole manifestations of cranial lesion

Authors:

Dmitri Pchejetski (d.pshezhetskiy@uea.ac.uk)
Mojiba Kenbaz (Mojiba.Kenbaz@jpaget.nhs.uk)
Heba Alshaker (H.Alshaker@uea.ac.uk)
Kiruparan Jesudason (Kiruparajan.Jesudason@jpaget.nhs.uk)

Version: 2 Date: 08 Nov 2019

Author’s response to reviews:

We thank the Reviewer for the time taken to evaluate our work and the insightful comments.

1. The mentioned abbreviations have been expanded in the manuscript.
2. Normal blood count, urinalysis and CXR have ruled out significant infection. We have amended manuscripts accordingly.